

Author's response to reviews

Title: Lewy bodies and neuronal loss in subcortical areas and disability in non-demented older people: a population based neuropathological cohort study

Authors:

Michelle Byford (michelle.byford@cantab.net)

Carol Brayne (carol.brayne@phpc.cam.ac.uk)

Ian McKeith (i.g.mckeith@newcastle.ac.uk)

Mark Chatfield (Mark.Chatfield@mrc-hnr.cam.ac.uk)

Paul G Ince (P.G.Ince@sheffield.ac.uk)

Fiona E Matthews (Fiona.matthews@mrc-bsu.cam.ac.uk)

MRC CFAS Neuropathology Group (Fiona.matthews@mrc-bsu.cam.ac.uk)

Version: 4 **Date:** 29 May 2009

Author's response to reviews: see over

Formatting changes requested

Major revisions (We require the author to make these changes)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Done, but note that the final author is corporate authorship and is actually “MRC CFAS Neuropathology group”, this is difficult to do on the online bit of the site.

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

Done, but note that the MRC Biostatistics Unit is the department and the institution.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.

Done

Reference 3 ζ please provide full and accurate details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city.

Done

Figure title - The image file should not include the title or figure number (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded. Please also remove the footnote.

Done

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].ζ

Done, I have now used the correct style. I have checked as far as I can without printing the references and apart from references that need to be in the middle of sentences or before a comma I could not find any that were not before the full-stop. Apologies if there are some, but I am in an airport and it isn't the easiest of things to do!

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Studyζ. To `The title of my prestigious publication: a scientific studyζ

Done

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system

Done, I could not find them in the submission section, but have removed them from the manuscript.

Please place the figure legend between the reference list and the Tables.

Done

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: <http://www.biomedcentral.com/info/ifora/figures>

Done.

The additional file is an update of the manuscript. Please upload this as the latest version if it is and remove the additional file.

Done, it was a version with tracked changes now just the clean version is on the system

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Thank you

To submit your revised manuscript

1. Go to 'my BioMed Central' (<http://www.biomedcentral.com/my/>) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.
2. Click on the 'Submit revision' button next to the title of this manuscript to go through to a submission form.
3. With the 'Manuscript details' tab, please update the title, abstract and author details if they have changed since the previous version. It is very important that all changes are updated on this page, as well as in the manuscript file as the information on this page will be used in PubMed and on BioMed Central.
4. With the 'Cover letter' tab, please provide a covering letter with a point-by-point description of the changes made.
5. With the 'Upload files' tab, please upload the revised version of the manuscript and press 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments. Please do not hesitate to contact us if you have any problems or questions.

Please note that this is the proofing stage. All revised manuscripts submitted to BMC Geriatrics are regarded as being in the final version. You will not be able to make any corrections to your article once it has been accepted and published in preliminary form. Therefore, when you submit your final version, please carefully check your manuscript for any typing or content errors that might have inadvertently occurred, and for small changes to the text that you or your co-authors wish to make at this time. This will be the last chance you have to make any such corrections as any errors present in the manuscript may end up in the final published version.

Please ensure that all revisions are made to the most recent version of your article which can be accessed at the following link:

<http://www.biomedcentral.com/media/man/3876617522419700/msd/main/0>

Figures:

<http://www.biomedcentral.com/media/man/3876617522419700/msd/FIG/1>

We would ask that you return any revisions by Saturday, 30 May 2009.

With best wishes,